Author's response to reviews

Title: Body weight, weight perceptions and food intake patterns. A cross-sectional study among male recruits in the Norwegian National Guard.

Authors:

Solveig Uglem (solveig.uglem@medisin.uio.no)
Tonje H Stea (tonje.h.stea@ui.no)
Wenche Frølich (wenche.frølich@live.no)
Margareta Wandel (margareta.wandel@medisin.uio.no)

Version: 4 Date: 4 May 2011

Author's response to reviews: see over
Dear Editor of BMC Public Health,

Thank you for the e-mail of 4 May informing me that some additional revisions needed to be made on our manuscript “Body weight, weight perceptions and food intake patterns; A cross-sectional study among male recruits in the Norwegian National Guard”, MS ID 3777765348694449. This is how we have revised the paper:

1. Author list.
   The author list is now amended so that is exactly the same as in the submission system.

2. Abstract.
   The abstract is now changed so that it is exactly the same as in the submission system.

3. Tables.
   All the tables are now created using the Tables tool.

   The tracking is now removed from the manuscript.

5. Typography.
   I have proofread the manuscript for any typographical errors, and made small corrections where needed.

Thank you for timely and professional handling in the publication of this manuscript

Kind regards

Margareta Wandel